Author’s response to reviews

Title: Can an Insole for Obese Individuals Maintain the Arch of the Foot Against Repeated Hyper Loading?

Authors:

Yuki Saito (yuki.saito@sapmed.ac.jp)
Takako Chikenji (chikenji@sapmed.ac.jp)
Yuichi Takata (takata@do-bunkyodai.ac.jp)
Tomoaki Kamiya (tkamiya0606@gmail.com)
Eiichi Uchiyama (uchiyama@sapmed.ac.jp)

Version: 2 Date: 05 Sep 2019

Author’s response to reviews:

Editor Comments:

Congratulations on a good response to review. Your manuscript has been accepted however please still make sure you make the minor changes highlighted by Dr Bruening.

Authors response: Thank you for your decision. We have revised our manuscript according to the suggestion from Dr. Bruening.

Dustin Bruening, Ph.D. (Reviewer 2): The authors have done an admirable job of responding to the critiques and modifying the manuscript accordingly.

Just a couple of minor grammatical suggestions:

add "was" to page 3, line 47: ...experimental model "was" designed...

add "commercial" to page 6 line 113: analyzed using "commercial" software...

Authors response: Thank you for your comment and suggestion. We have corrected the two grammatical error according to your suggestions.